Author’s response to reviews

Title: Requiring smartphone ownership for mHealth interventions: Who could be left out?

Authors:
Krishna K. Bommakanti (kbommakanti@ucsd.edu)
Laramie L. Smith (laramie@ucsd.edu)
Lin Liu (l2liu@ucsd.edu)
Diana Do (dianado26@yahoo.com)
Jazmine Cuevas-Mota (jcuevasmota@ucsd.edu)
Kelly Collins (Kelly.Collins@sureadhere.com)
Fatima Munoz (fmunoz987@gmail.com)
Timothy C. Rodwell (trodwell@ucsd.edu)
Richard S Garfein (rgarfein@ucsd.edu)

Version: 3 Date: 31 Oct 2019

Author’s response to reviews:

Thank you for amending your manuscript according to our requests. However, there are a few minor revisions that are necessary before we can accept the manuscript for publication:1. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations. We have added a section for abbreviations in the 'Declarations' section. All abbreviations used in the text are defined at first use and included in the abbreviations section.